Author's response to reviews

Title: Comparison of Quality-of-Care Measures in U.S. Patients with End-Stage Renal Disease Secondary to Lupus Nephritis vs. Other Causes

Authors:

Laura C Plantinga (laura.plantinga@emory.edu)
Rachel E Patzer (rpatzer@emory.edu)
Cristina Drenkard (cdrenka@emory.edu)
Stephen O Pastan (Stephen.Pastan@emoryhealthcare.org)
Jason Cobb (jcobb2@emory.edu)
William McClellan (wmcclel@emory.edu)
S. Sam Lim (sslim@emory.edu)

Version: 4  Date: 14 March 2015

Author's response to reviews: see over
Dear Dr. Weisbord:

We are pleased to submit a final revision of our original article, entitled “Comparison of Quality-of-Care Measures in Patients with End-Stage Renal Disease” (manuscript no. 818020291525330). We have addressed the editorial comments as follows:

Editorial Requests:
1. Check the Authors’ Contributions - currently 8 authors are mentioned, are LCP and LP referring to the same person?

Response. Yes, thank you. LCP and LP are the same author (i.e., only 7 authors are mentioned) and this has been corrected in the Authors’ Contributions.

2. Check formatting - You now have an opportunity to check that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.

Response. We have formatted our paper per the journal style.

We appreciate the opportunity publish our work in BMC Nephrology.

Sincerely,
Laura Plantinga, PhD